\documentclass[12pt]{umj_eng3}
\usepackage[english]{babel}
\usepackage{amsmath}
\usepackage{amssymb}

\newtheorem{lemma}{Lemma}[section]
\newtheorem{theorem}{Theorem}[section]
\newtheorem{definition}{Definition}[section]



\def\udcs{517.956} %Çäåñü àâòîð îïðåäåëÿåò ÓÄÊ ñâîåé ðàáîòû

\def\currentyear{2026}
\def\currentvolume{18}
\def\currentissue{2}
\lastpage{141}
\urle{umj.ufaras.ru/index.php/umj/article/view/2026-18-2-131}
\renewcommand{\baselinestretch}{1.00}
\firstpage{131}


\renewcommand{\leq}{\leqslant}
\renewcommand{\geq}{\geqslant}
\renewcommand{\le}{\leqslant}
\renewcommand{\ge}{\geqslant}

%%%%% àâòîðñêèå ìàêðîñû %%%%%

\def\tht{\theta}
\def\Om{\Omega}
\def\om{\omega}
\def\e{\varepsilon}
\def\g{\gamma}
\def\G{\Gamma}
\def\l{\lambda}
\def\p{\partial}
\def\D{\Delta}
\def\bs{\backslash}
\def\k{\varkappa}
\def\E{\mbox{\rm e}}
\def\a{\alpha}
\def\b{\beta}

\def\si{\sigma}
\def\Si{\Sigma}
\def\d{\delta}
\def\L{\Lambda}
\def\z{\zeta}
\def\vs{\varsigma}
\def\r{\rho}
\def\vp{\varphi}
\def\vw{\varpi}
\def\vr{\varrho}
\def\vt{\vartheta}
\def\h{\widehat}
\def\Odr{O}
\def\H{W_2}
\def\Ho{\mathring{W}_2}
\def\Hloc{W_{2,loc}}
\def\Hper{W_{2,per}}
\def\Hoper{\mathring{W}_{2,per}}
\def\Hinf{W_\infty}

\def\mD{{D}}
\def\mR{{R}}


\def\Th{\Theta}
\def\I{\mathrm{I}}
\def\iu{\mathrm{i}}
\def\di{\,d}



\def\cH{\mathcal{H}}
\def\cS{\mathcal{S}}

\def\la{\langle}
\def\ra{\rangle}

\def\cB{\mathcal{B}}
\def\cP{\mathcal{P}}
\def\cI{\mathcal{I}}
\def\cL{\mathcal{L}}
\def\cA{\mathcal{A}}
\def\cM{\mathcal{M}}
\def\cQ{\mathcal{Q}}

%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\DeclareMathOperator{\RE}{Re}
\DeclareMathOperator{\IM}{Im}
\DeclareMathOperator{\spec}{\sigma}
\DeclareMathOperator{\discspec}{\sigma_{disc}}
\DeclareMathOperator{\essspec}{\sigma_{ess}}
\DeclareMathOperator{\dist}{dist}
\DeclareMathOperator{\supp}{supp}
\DeclareMathOperator{\sgn}{sgn}
\DeclareMathOperator{\rank}{rank}
\DeclareMathOperator*{\essinf}{ess\,inf}
\DeclareMathOperator{\rot}{rot}
\DeclareMathOperator{\dvr}{div}
\DeclareMathOperator*{\esssup}{ess\,sup}

\DeclareMathOperator{\Dom}{\mathfrak{D}}


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%

\numberwithin{equation}{section}

\begin{document}
\begin{thebibliography}{999}



\bibitem{Aloui2013JMPA}
F. Aloui, I.B. Hassen, A. Haraux. {\it Compactness of trajectories to some nonlinear second
order evolution equations and applications} //  J. Math. Pures Appl. {\bf 100}:3, 295--326 (2013).

\noindent\hrr{https://doi.org/10.1016/j.matpur.2013.01.002}


 \bibitem{Aljurbua2025}
S. F. Aljurbua, A. Bchatnia, A. Beniani,  A. Safsaf. {\it   Blow-up for coupled nonlinear wave equations with fractional damping and logarithmic source terms} //  Comput. Appl. Math. {\bf 44}:8, 400 (2025).

\noindent\hrr{https://doi.org/10.1007/s40314-025-03362-0}

 \bibitem{Balakrishnan1989}
A.V. Balakrishnan and  L.W. Taylor. {\it  Distributed parameter nonlinear damping models for flight structures} // in Proceedings ``Damping 89'', Flight Dynamics Lab and Air Force Wright
Aeronautical Labs, WPAFB  (1989).

\bibitem{Chueshov2008AMS}
I. Chueshov and I. Lasiecka. {\it  Long-Time Behavior  of Second Order Evolution Equations with Nonlinear Damping}. Amer. Math. Soc., Providence, Rhode Island (2008).

\noindent
 \hrr{https://doi.org/10.1090/memo/0912}



\bibitem{Ghisi2019ANA}
M. Ghisi, M. Gobbino and A. Haraux. {\it  Quantization of energy and weakly turbulent profiles of solutions to some damped second--order evolution equations} //
Adv. Nonlinear Anal. {\bf 8}, 902--927 (2019).
 \hrr{https://doi.org/10.1515/anona-2017-0181}



\bibitem{Jorgens1961}
 K. J\"orgens. {\it  Des aufangswert in grossen f$\ddot{u}$r eine klasse nichtlinearer
 wallengleinchungen} //  Math. Z. {\bf 77}:1, 295--308 (1961). \hrr{https://doi.org/10.1007/BF01180181}



 \bibitem{Silva2019DCDS}
 M.A.~Jorge Silva, V.~Narciso, and A.~Vicente. {\it
 On a beam model related to flight structures
with nonlocal energy damping} //
Discrete Contin. Dyn. Syst., Ser. B {\bf24}:7, 3281--3298 (2019).


\noindent\hrr{https://doi.org/10.3934/dcdsb.2018320}

\bibitem{Zhang2023QM}
Q. Y. Hu, D. H. Li, S. Liu, H. W. Zhang. {\it
Blow-up of solutions for a wave equation with nonlinear
averaged damping and nonlocal nonlinear source terms} //
Quaest. Math. {\bf 46}:4, 695--710 (2023).


\noindent\hrr{https://doi.org/10.2989/16073606.2022.2039972}


\bibitem{Zhang2019JPDE}
D. H. Li, H. W. Zhang, Q. Y. Hu. {\it
General energy decay of solutions for
a wave equation with nonlocal damping and nonlinear boundary damping} //
J. Part. Diff. Equ. {\bf 32}:4,  369--380 (2019).


\noindent\hrr{https://doi.org/10.4208/jpde.v32.n4.6}

\bibitem{Li2022EECT}
D.H.~Li, H.W.~Zhang, S.~Liu, and Q.Y.~Hu. {\it
Blow-up of solutions to a viscoelastic wave equation with nonlocal damping} //
Evol. Equ. Control Theory {\bf11}:6, 2017--2031 (2022).


\noindent\hrr{https://doi.org/10.3934/eect.2022009}

\bibitem{Louredo2014}
A. T. Lour\^edo, M. A. Ferreira de Ara\'ujo, M. M. Miranda. {\it  On a
nonlinear wave equation with boundary damping}  // Math. Meth. Appl. Sci. {\bf 37}:9, 1278--1302 (2014).


\noindent\hrr{https://doi.org/10.1002/mma.2885}

\bibitem{Messaoudi}
 S. A. Messaoudi. {\it  Global existence and nonexistence in a system of Petrovsky}  //  J. Math. Anal. Appl. {\bf 265}:2, 296--308 (2002). \hrr{https://doi.org/10.1006/jmaa.2001.7697}


\bibitem{Peng2023AMO}
Q.Q.~Peng and  Z.F.~Zhang. {\it  Global attractor for a coupled wave and plate equation
with nonlocal weak damping on Riemannian manifolds} //
Appl. Math. Optim. {\bf 88}:2, 28 (2023).

\noindent\hrr{https://doi.org/10.1007/s00245-023-09998-w}

\bibitem{Schiff1951}
L. Schiff. {\it Nonlinear meson theory of nuclear forces I. Neutral scalar mesons
with point-contact repulsion} //  Phys. Reviews. {\bf 84}:1, 1--9 (1951). \hrr{https://doi.org/10.1103/PhysRev.84.1}

\bibitem{Yang2022JMAA}
Y. Sun,  Z. J.Yang. {\it   Attractors and their continuity for an extensible
beam equation with rotational inertia and nonlocal energy damping} //
J. Math. Anal. Appl. {\bf512}:2, 126148 (2022).


\noindent\hrr{https://doi.org/10.1016/j.jmaa.2022.126148}


\bibitem{Wu2009TJM}
S.T.~ Wu and  L.Y.~ Tsai. {\it  On global solutions and blow--up of solutions for a nonlinearly damped Petrovsky system} // Taiwanese J. Math. {\bf 19}:2A, 545--558 (2009).


\noindent \hrr{https://doi.org/10.11650/twjm/1500405355}

\bibitem{zennir2018}
Kh. Zennir, M. Bayoud, and S. Georgiev. {\it  Decay of solution for degenerate wave equation of Kirchhoff type in viscoelasticity} //   Int. J. Appl. Comput. Math. {\bf 4}:1, 54 (2018).


\noindent\hrr{https://doi.org/10.1007/s40819-018-0488-8}

\bibitem{Zennir2015}
Kh. Zennir, A. Guesmia. {\it  Existence of solutions to nonlinear $\kappa$th-order
coupled Klein~---~Gordon equations with nonlinear sources and memory terms} //   Appl. Math. E-Notes. {\bf 15}, 121--136 (2015).

\noindent\hrr{https://emis.de/ft/8211}

\bibitem{Zennir2024} H. Zhang, L. Liu, H. Yue, D. Li, Kh. Zennir. {\it  The existence, asymptotic behaviour and blow--up of solution of a plate equation with nonlinear averaged damping} //   Rep. Math. Phys. {\bf  19}:3, 305--323 (2024). \hrr{https://doi.org/10.1016/S0034-4877(24)00081-8}

\bibitem{Zhang2022AA}
H.W.~Zhang, D.H.~Li, W.X.~Zhang, and Q.Y.~Hu. {\it
Asymptotic stability and blow--up for the wave equation with degenerate nonlocal
nonlinear damping and source terms} // Appl. Anal. {\bf 101}:9, 3170--3181 (2022). \hrr{https://doi.org/10.1080/00036811.2020.1836354}

\bibitem{Miyasita2020}
Kh. Zennir, T. Miyasita. {\it  Lifespan of solutions for a class of pseudo-parabolic equation with weak-memory} //  Alexandria Eng. J. {\bf59}:2, 957--964 (2020). \hrr{https://doi.org/10.1016/j.aej.2020.03.016}

\bibitem{Zhao2020DCDS}
C.Y. Zhao, C.X. Zhao,  C.K.Zhong. {\it The global attractor for a class of extensible beams with
nonlocal weak damping} //  Discrete Contin. Dyn. Syst., Ser. B {\bf25}:3, 935--955 (2020).


\noindent
\hrr{https://doi.org/10.3934/dcdsb.2019197}


\bibitem{Zhao2023ActaMS}
C.X. Zhao, F.J. Meng. {\it Attractor for the extensible beam equation with nonlocal weak
damping on time-dependent space} //  Acta Math. Sin., Engl. Ser. {\bf
40}:4, 1115--1126 (2024).


\noindent\hrr{https://doi.org/10.1007/s10114-023-1295-7}
\end{thebibliography}


%ÓÄÊ \udcs
 \thispagestyle{empty}


\title[Blow--up for system of $\kappa$th order wave equations]{Finite time blow--up  of solutions for system of $\kappa$th order wave equations with nonlinear averaged damping and sources}


\author{K\lowercase{h}. Zennir, K. Bouhali, H. Zhang and D. Guidad}

\address{Khaled Zennir,
\newline\hphantom{iii} Department of Mathematics,
\newline\hphantom{iii} College of Science,
\newline\hphantom{iii} Qassim University,
\newline\hphantom{iii} Buraidah, Saudi Arabia.
}
\email{k.zennir@qu.edu.sa}

\address{Keltoum Bouhali,
\newline\hphantom{iii} Department of Mathematics,
\newline\hphantom{iii} College of Science,
\newline\hphantom{iii} Qassim University,
\newline\hphantom{iii} Buraidah, Saudi Arabia.
}
\email{k.bouhali@qu.edu.sa}


\address{Hongwei Zhang,
\newline\hphantom{iii} School of Mathematics and Statistics,
\newline\hphantom{iii} Henan
University of  Technology,
\newline\hphantom{iii} Zhengzhou 450001, China.
}
\email{whz661@163.com}

\address{Derradji Guidad,
\newline\hphantom{iii} Department of Mathematics,
\newline\hphantom{iii} College of Sciences,
\newline\hphantom{iii} University of Biskra,
\newline\hphantom{iii} BP 145 RP, 07000 Biskra,  Algeria.}
\email{guidadd07@gmail.com}


\thanks{\textsc{Kh. Zennir, K. Bouhali, H. Zhang and D. Guidad, Finite time blow--up  of solutions for   %nonlinear coupled
system of $\kappa$th order wave equations with nonlinear averaged damping and sources}}
\thanks{\textcopyright\ Zennir Kh., Bouhali K., Zhang H. and Guidad D.  2026}
\thanks{\it Submitted February 27, 2025.}

\maketitle {\small
\begin{quote}
\noindent\noindent{\bf Abstract.}   We consider basic relationships between the initial energy and the exponent of nonlinear sources in a nonlinear coupled system of $\kappa$th order wave equations with nonlinear averaged damping, and we demonstrate the global nonexistence of solutions. This approach is a variant of the method of nonlinear functional analysis  with a contradiction argument, which is one of the tools for proving the blow--up of solutions for nonlinear partial differential equations.    A new class of nonlinear coupled wave equations with nonlinear sources is given in high-order functional spaces.   %The developed approach allows for blow--up results.




\medskip



\noindent{\bf Keywords:} $\kappa$th order wave equations, nonlinear equations, coupled system, nonlinear averaged damping, local existence,  blow--up.

\medskip
\noindent{\bf Mathematics Subject Classification: }{35A01; 35L20; 35B44}

\end{quote}
}


\vskip30pt

\section{Introduction, relevance and objectives }
Let $\Omega \subset \mathbb{R}^n$ $(n\ge 1)$ be a bounded domain with a smooth boundary $\Gamma$.
%Let $u=u(x, t), v=v(x, t)$, for $x\in \Omega, t \in [0, +\infty)$,
We consider the initial boundary value problem for nonlinear $\kappa$th order wave equations
with a\ nonlocal damping
\begin{equation}\label{1.1}
\left\{
\begin{aligned}
&u_{tt}+ (-\Delta)^\kappa u
+\|u_t\|^{l}u_t=|u|^{p-2}u\vert v\vert ^p,
\\
&v_{tt}+ (-\Delta)^\kappa v
+\|v_t\|^{l}v_t=|v|^{p-2}v\vert u\vert ^p,
\end{aligned}
\right. \qquad x\in\Om,\quad t>0,
\end{equation}
with boundary conditions
\begin{equation}  \label{1.2}
\left\{
\begin{aligned}
u=\frac{\partial u}{\partial \nu}=\ldots=\frac{\partial^{\kappa -1} u}{\partial \nu^{\kappa -1}}=0,
 \\
v=\frac{\partial v}{\partial \nu}=\ldots=\frac{\partial^{\kappa -1} v}{\partial \nu^{\kappa -1}}=0,
\end{aligned}
\right.\qquad x\in  \Gamma,
\end{equation}
and initial conditions
\begin{equation} \label{1.3}
\left\{
\begin{aligned}
&u(x,0) = u_0(x),\qquad u_t(x,0) = u_1(x),
\\
&v(x,0) = v_0(x),\hspace{3 true pt}\qquad v_t(x,0) = v_1(x),
\end{aligned}
\right.\qquad x\in\Om.
\end{equation}
Here $\nu $ is the outward normal to the boundary $\G,$ while  $l>0,$ $\kappa\geq 1$ are integers and $p>2$, and  for $u=u(x, t)$
\begin{align*}
&\vert\nabla^{\kappa}u\vert^{2}=(\Delta^{\frac{\kappa}{2}}u)^{2}\hspace{26 true pt}\qquad \text{for even}\quad  \kappa,
\\
&\vert\nabla^{\kappa}u\vert^{2}=\vert\nabla(\Delta^{\frac{\kappa-1}{2}}u ) \vert^{2}
\qquad \text{for even}\quad   \kappa,
\end{align*}
and
\begin{equation*}
\vert\nabla u\vert^{2}= \sum_{i=1}^{n}\left(\frac{\partial u}{\partial x_{i}}\right)^{2},
\qquad \Delta u = \sum_{i=1}^{n}\frac{\partial^{2} u}{\partial x_{i}^{2}}.
\end{equation*}
Recently, an initial boundary value problem for following wave equation with nonlinear averaged damping
\begin{align}
&u_{tt}- \Delta  u+g\Big(\int\limits_{\Omega}|u_t|^2\,dx\Big)u_t=f(u) \label{1.4}
\end{align}
attracted a lot of attention. Equation \eqref{1.4} with the source $f(w)=w\int\limits_{\Omega}|w|^2\,dx$ arises from a nonlinear theory of meson fields \cite{Jorgens1961}, \cite{Louredo2014}, \cite{Schiff1951}.  It can also be intrinsically connected with the
study of damping phenomena in flight structures, where the   $\|u_t \|^m$
reflects the effect of kinetic energy; see   \cite{Aljurbua2025}, \cite{Silva2019DCDS}, \cite{Zennir2024}.
For instance, in  \cite{Balakrishnan1989}, the   one dimensional beam equation
\begin{align}
 u_{tt}-2\varsigma\sqrt{\lambda}u_{xx}+\lambda u_{xxxx}
-\gamma\Bigg(\int\limits_{L}^{-L}(\lambda| u_{xx}|^2+| u_{t}|^2\,dx\Bigg)^q u_{xxt}=0 \label{1.5}
\end{align}
was proposed.
As the first stage of the corresponding equation
$$u_{tt}+\Big(\int\limits_\Omega u_t^2\,dx\Big)^\frac{\alpha}{2}u_t-\Delta u=h(t,x),$$
the pioneering works \cite{Aloui2013JMPA} provided the above equation in a bounded domain $\Omega$
with homogeneous boundary conditions; see  \cite{Aloui2013JMPA}, \cite{Ghisi2019ANA}, \cite{zennir2018}. Later, the authors in \cite{Louredo2014} studied the system
\begin{equation}
\begin{aligned}
&u_{tt}- \mu(t)\Delta u+\alpha f\bigg(\int\limits_\Omega |u|^{2}\,dx\bigg)u+\beta g\bigg(\int\limits_\Omega |u_t|^{2}\,dx\bigg)u_t
=0,  \\
&u=0\quad\text{on}\quad \Gamma_0,
\qquad \frac{\partial u}{\partial \nu} +h(\,\cdot\,,u_t)=0\quad\text{on}\quad \Gamma_1,  \\
&u(x, 0) = u_0(x),\qquad u_t(x,0) = u_1(x),
\end{aligned}
\end{equation}
under certain conditions on $\mu(t),$ $ f(s), $ $ g(s), $ $ h(s)$. Global
existence in time was shown by using the well--known Galerkin method. In addition,  a decay rate of the solution was obtained by two different methods, both Lyapunov functional and Nakao's lemma.
Zhang \cite{Zhang2019JPDE} extended and improved
the result in \cite{Louredo2014}. In \cite{Zhao2020DCDS}, \cite{Zhao2023ActaMS}, the long time behavior of the  equations
\begin{align}
&u_{tt}+\Delta ^2u-M(\|\nabla u\|^{2}) \Delta u+k(\|u_t\|)u_t+f(u)=h(x) \label{1.10}
\\
&u_{tt}-\Delta u+k(\|u_t\|)u_t+f(u)=\int\limits_{\Omega}K(x,y)u_t(y)dy+h(x) \label{1.11}
\end{align}
was considered.
The global well--posedness of \eqref{1.10} \eqref{1.11} was established  with the  damping $\|u_t\|^m u_t,$ $m\ge 0,$  see
\cite{Li2022EECT}, \cite{Peng2023AMO}, \cite{Zhang2022AA} and references therein. In \cite{Silva2019DCDS}, the authors investigated the nonhomogeneous $n$--dimensional version of \eqref{1.5}
\begin{equation}
{u_{tt}} - \kappa{\Delta }u +{\Delta^2 }u-  \gamma \bigg(\int\limits_\Omega  {({{\left| {\Delta u} \right|}^2} + {{\left| {{u_t}} \right|}^2})\,dx} \bigg)^q\Delta {u_t}+f(u) =0  \quad\text{in}\quad \Omega\times {R}^+ \label{1.12}
\end{equation}
 subject to a boundary condition and with $q\geq1$. The authors proved the existence of  a unique global solution and obtained the polynomial stability and a non-exponential decay prospect of the
solutions provided that $q\geq1,$ see  \cite{Chueshov2008AMS}, \cite{Yang2022JMAA}, \cite{Zennir2015}.


 For the equation
\begin{align}
&u_{tt}- \Delta u+ \bigg(\int\limits_\Omega |u_t|^{2}\,dx\bigg)^\frac{m}{2}u_t= \bigg(\int\limits_\Omega |u|^{p}\,dx\bigg)^r|u|^{p-2}u \label{1.13}
\end{align}
with homogeneous boundary conditions, Hu, Li, Liu and Zhang  \cite{Zhang2023QM}  provided
sufficient conditions for finite time blow--up of weak solutions, with suitable conditions
on initial data, by an ordinary differential inequality for an appropriately chosen functional.
When the terms $\big(\int\limits_\Omega |u|^{p}\,dx\big)^r|u|^{p-2}u$ and $-\Delta u$ in equation \eqref{1.13}
are replaced by $|u|^{p-2}u $ and $ \Delta^2 u$, respectively,
 Zennir and Miyasita \cite{Miyasita2020} proved the global non-existence in time for a large class of pseudo-parabolic equations with weak viscoelasticity under suitable conditions on the variable exponents with negative initial energy.


To our knowledge, there are few results on the non-linear coupled system of $\kappa$th order wave equations with nonlinear averaged damping and sources. The main goal of this paper is to study the blow--up conditions of a solution to problem \eqref{1.1}--\eqref{1.3} with the averaged damping term and the source term, which differ from the existing literature mentioned above.  To the best of our knowledge, this is the first result on the nonlinear coupled system of $\kappa$th order
wave equations with nonlinear averaged damping and sources.


The paper is organized as follows. Section 2 is concerned with some notation and necessary materials.  In Section 3, we demonstrate the global nonexistence of solutions with positive or negative initial energy.

\vskip20pt

\section{Preliminaries}

Let  $L^2(\Omega),H^p(\Omega)$ and $H^2_0(\Omega)$ be the usual Sobolev spaces, where
\begin{equation*}
(u,v)=\int\limits_{\Omega}uv\,dx, \qquad \|u\|=\bigg(\int\limits_{\Omega}|u|^2\,dx\bigg)^\frac{1}{2},\qquad
\|u\|_p
=\bigg(\int\limits_{\Omega}|u|^p\,dx\bigg)^\frac{1}{p}.
\end{equation*}

\begin{definition}
Functions $$(u, v)\in  C\Big([0, T]; H ^\kappa(\Omega)\Big) \cap C^1 \Big([0, T]; L^2 (\Omega)\Big)\times  C\Big([0, T]; H ^\kappa(\Omega)\Big) \cap C^1 \Big([0, T]; L^2 (\Omega)\Big),$$
with the initial data given in \eqref{1.3} are

\vskip5pt

\begin{enumerate}
    \item  a strong solution to problem \eqref{1.1}--\eqref{1.3} on the interval $[0, T]$
 if
 \begin{align*}
 &(u,v) \in W^{1,1}(\alpha, \beta; H^\kappa_0(\Omega)) \times W^{1,1}(\alpha, \beta; H^\kappa_0(\Omega)),
 \\
 &(u_t, v_t)\in W^{1,1}(\alpha, \beta; L^2(\Omega)) \times W^{1,1}(\alpha, \beta; L^2(\Omega)),
 \end{align*}
 for all $0 < \alpha < \beta < T$ and $$\Delta^\kappa u + \Delta^\kappa v + \|u_t(t)\|^l u_t + \|v_t(t)\|^l v_t \in L^2(\Omega)\quad \text{for a.e.}\quad t\in [0,T],$$
  and \eqref{1.1} is satisfied in $L^2 (\Omega)$ for all $t\in [0, T];$

\vskip5pt

\item  a generalized solution to \eqref{1.1}-\eqref{1.3} on the interval $[0, T]$
  if there is a sequence of strong solutions $(\{u_j (t)\}, \{v_j (t)\})$ to \eqref{1.1}-\eqref{1.3} with initial data $(u^j_0, v^j_0), (u^j_1, v^j_1)$ instead of $(u_0, v_0 ), (u_1, v_1 )$
such that
\begin{align*}
&(u^j, v^j )\rightarrow  (u, v)\quad \hspace{6.5 true pt}\text{in}\quad C([0, T]; H^\kappa_0(\Omega))\times C([0, T]; H^\kappa_0(\Omega)) \quad \text{as} \quad j \rightarrow +\infty,
\\
&(u^j_t, v^j_t)\rightarrow  (u_t, v_t ) \quad \text{in}\quad C([0, T]; L^2(\Omega))\times C([0, T]; L^2(\Omega))\hspace{6.5 true pt}\quad \text{as} \quad j \rightarrow +\infty.
\end{align*}

\vskip5pt

\item  a distributed solution to \eqref{1.1}--\eqref{1.3} on $[0, T]$  if
\begin{equation}
\begin{aligned}
\int\limits_{\Omega}&\big(u_{t}(x,t)\phi (x)+v_{t}(x,t)\psi(x)\big)\,dx
\\
&+\int\limits_0^t\Bigg(\int\limits_{\Omega} \nabla^\kappa u(x,s)\nabla^\kappa \phi(x)\,dx
+\|u_t(s)\|^l\int\limits_{\Omega} u_t(x,s)\phi(x)\,dx\Bigg)\,ds
\\
&+\int\limits_0^t\Bigg(\int\limits_{\Omega} \nabla^\kappa v(x,s)\nabla^\kappa \psi(x)\,dx
+\|v_t(s)\|^l\int\limits_{\Omega} v_t(x,s)\psi(x)\,dx\Bigg)\,ds
\\
=&\int\limits_{\Omega}u_1\phi (x)\,dx+\int\limits_0^t\int\limits_{\Omega}|u|^{p-2}u(x,s)|v(x, s)|^p\phi(x)\,dxds
\\
&+\int\limits_{\Omega}v_1\psi (x)\,dx+\int\limits_0^t\int\limits_{\Omega}|v|^{p-2}v(x,s)|u(x, s)|^p\psi(x)\,dxds
\end{aligned}\label{2.1}
\end{equation}
holds for every test function $\phi, \psi \in H^\kappa_0(\Omega)\cap L^2(\Omega)$ and for all $t\in [0, T].$
\end{enumerate}
\end{definition}


\begin{lemma}\label{lemma2.2} \cite{Zhao2020DCDS} For $u_t, w_t\in L^2(\Omega)$, we have
\begin{align}
& \|u_t-w_t\|^{l+2} \leq \left(\|u_t\|^lu_t-\|w_t\|^lw_t,u_t-w_t\right).\label{2.2}
\end{align}
\end{lemma}
The associated energy of  \eqref{1.1}-\eqref{1.3} is defined  by
\begin{align}
&E(t)=\frac{1}{2}\|u_t\|^2+\frac{1}{2}\|v_t\|^2+J(u, v),\quad 0\le t,\label{2.3}\\
&J(u, v)=\frac{1}{2}\|\nabla^\kappa u\|^2+\frac{1}{2}\|\nabla^\kappa v\|^2-\frac{1}{p}\|uv\|_p^p,\label{2.4}
\end{align}
and
\begin{align}
&E|_{t=0}=\frac{1}{2}\|u_1\|^{2}+\frac{1}{2}\|v_1\|^{2}+J(u_0, v_0),\label{2.5}
\\
&\|u\|_p\le C_0\|\nabla^\kappa u\|, \ C_0>0, \ \forall u\in H_0^\kappa(\Omega),\nonumber
\end{align}
 where
\begin{equation}
\left\{
\begin{aligned}
&2 \leq p <+\infty && \text{if}\quad  n=\kappa,2\kappa
 \\
&2 \leq p \leq \frac{2n}{n-2\kappa} && \text{if}\quad n\geq 3\kappa.
\end{aligned}
\right.  \label{pcondition}
\end{equation}

\vskip20pt

\section{Finite time blow up of solutions}

We state the local existence in time of \eqref{1.1}--\eqref{1.3} by following the lines of
 \cite{Chueshov2008AMS}.

\begin{theorem} \label{Theoremlocal} If $(u_0, v_0) \in (H^{2\kappa}(\Omega) \cap H^\kappa_0(\Omega))\times (H^{2\kappa}(\Omega) \cap H^\kappa_0(\Omega))$
and $(u_1, v_1) \in H^\kappa_0(\Omega) \times H^\kappa_0(\Omega)$ are such
that $$\nabla^\kappa u_0 +\nabla^\kappa v_0 +k\|u_1 \|^l u_1+k\|v_1 \|^l v_1 \in L^2 (\Omega).$$
Suppose that $l> 0$ and $p$ satisfies
\begin{equation} \label{c3.1}
\left\{
\begin{aligned}
&1 \leq p <+\infty &&  \text{if}\quad n=\kappa,2\kappa  \\
&1 \leq p \leq \frac{4\kappa-n}{n-2\kappa}\quad &&  \text{if}\quad  n\geq 3\kappa.
\end{aligned}
\right.
\end{equation}
Then  there exists $t_{\mathrm{max}}\le +\infty$
such that  the problem  \eqref{1.1}--\eqref{1.3}
has a unique strong solution $(u, v)$
on $[0, t_{\mathrm{max}})$.


If $(u_0, v_0) \in H^\kappa_0(\Omega)\times H^\kappa_0(\Omega)$ and $(u_1, v_1) \in L^2(\Omega)\times L^2(\Omega)$ ,
then there exists $t_{\mathrm{max}}\le +\infty$
such that \eqref{1.1}--\eqref{1.3} admits a
unique generalized solution,
which is also the distributional solution.


With positive and negative initial data, both the strong solution and the weak solution $(u, v)$
 cannot be global solutions in $[0, t_{max} ), \ t_{\mathrm{max}} = +\infty$,   so for
$t_{\mathrm{max}} < \infty$ we have
\begin{align}
&\lim_{t\rightarrow t_{\mathrm{max}}}\left(\|\nabla^\kappa u\|+\|\nabla^\kappa v\|+\|u_t\|+\|v_t\|\right)=+\infty,\label{3.1}
\end{align}
 The energy relation is satisfied
\begin{align}
&E(t)+\int\limits_0^t(\|u_t(s)\|^{l+2}+\|v_t(s)\|^{l+2})ds=E(0),\quad 0\le t<t_{\mathrm{max}}.\label{3.2}
\end{align}
\end{theorem}

\vskip10pt

\subsection{Negative initial energy} Our first main result is as follows.

\begin{theorem} \label{theoremblowup}
Let $E(0) < 0$. Under the assumptions of Theorem~\ref{Theoremlocal} there exists a finite time $t_{\mathrm{max}}<+\infty,$ at which the local solution to \eqref{1.1}--\eqref{1.3} blows up.
\end{theorem}

\begin{proof}
Let $$\mathcal{K}(t)=-E(t).$$ The contradiction argument will be used. By \eqref{2.4} and \eqref{3.2}  we obtain
\begin{align}
& \mathcal{K}'(t)=-E'(t)\ge 0, \nonumber
\\
&0<- E(0)= \mathcal{K}(t=0)\le \mathcal{K}(t)\le \frac{1}{p}\|uv\|^p_p.\label{5.3}
\end{align}
We denote
$$G(t)=\int\limits_{\Omega}(\|u_t\|^lu_tu+\|v_t\|^lv_tv)\,dx,$$
then, by H\"older and $\varpi$-Young's inequalities for each $\varpi>0$ we obtain
\begin{align*}
&
\|u_t\|^2\,\|u\|^2
\le \varepsilon \|u_t\|^{2} + \frac{1}{4\varepsilon}\|u\|^{2},
\\
&|G(t)|\le \|u_t\|^l\|u_t\|\|u\|+\|v_t\|^l\|v_t\|\|v\|
\\
&\hphantom{|G(t)|}\leq \|u_t\|^l\left(  \varpi \|u_t\|^2+\frac{1}{4\varpi} \|u\|^2\right) +  \|v_t\|^l\left( \varpi  \|v_t\|^2+\frac{1}{4\varpi}  \|v\|^2\right)
\\
&\hphantom{|G(t)|}\leq \varpi[\|u_t\|^{l+2}+\|v_t\|^{l+2}]+ \frac{1}{4\varpi}  [\|u_t\|^l\|u\|^2 +\|v_t\|^l\|v\|^2]
\end{align*}
Again, using Young's inequality with
$$
p=\frac{l+2}{l}, \qquad q=\frac{l+2}{2}, \qquad \frac{1}{p}+\frac{1}{q}=1,
$$
for each  $\varpi>0$ we obtain,
\begin{equation*}
\|u_t\|^{l}\|u\|^{2}\leq  \varpi\|u_t\|^{lp}
+ \varpi^{-\frac{2}{l}}\|u\|^{2q} =  \varpi \|u_t\|^{l+2}
+ \varpi^{-\frac{2}{l}}\|u\|^{l+2},
\end{equation*}
and similarly,
$$
\|v_t\|^{l}\|v\|^{2}
\leq \varpi \|v_t\|^{l+2}
+ \varpi^{-\frac{2}{l}}\|v\|^{l+2}.
$$
Hence,
 $$
 \Vert u_t\Vert ^l \Vert u\Vert ^2+ \Vert v_t\Vert ^l \Vert v\Vert ^2 \leq \varpi (\Vert u_t\Vert ^{l+2} +\Vert v_t\Vert ^{l+2} )+\varpi^{-\frac{2}{l}}(\Vert u\Vert ^{l+2} +\Vert v\Vert ^{l+2} ),
 $$
 which leads by \eqref{3.2} to
\begin{equation} \label{5.5}
\begin{aligned}
 |G(t)| \leq&  (\varpi+\frac{1}{4})[\|u_t\|^{l+2}+\|v_t\|^{l+2}]+\frac{ \varpi^{-\frac{2+l}{l}}}{4 }(\Vert u\Vert ^{l+2} +\Vert v\Vert ^{l+2} )  \\
 \leq& (\varpi+\frac{1}{4})\mathcal{K}'(t) +\frac{\varpi^{-\frac{2+l}{l}}}{4} (\Vert u\Vert ^{l+2} +\Vert v\Vert ^{l+2} ).
 \end{aligned}
\end{equation}
We define the functional
\begin{align}
&z(t)=\mathcal{K}^{1-\nu}(t)+\epsilon \int\limits_{\Omega} u_{t}u\,dx+v_{t}v\,dx, \qquad \epsilon>0,\qquad 0<\nu<1. \label{5.6}
\end{align}
Taking the derivative of \eqref{5.6} in $t$,
by \eqref{1.1}, and \eqref{5.5}, we have
\begin{equation}
\begin{aligned}
 z'(t) =& (1-\nu)\mathcal{K}^{-\nu}(t)\mathcal{K}'(t)
+\epsilon \int\limits_{\Omega} (u_{tt}u+v_{tt}v)\,dx+\epsilon (\|u_t\|^2+ \|v_t\|^2)
\\
 =& (1-\nu)\mathcal{K}^{-\nu}(t)\mathcal{K}'(t)
+\epsilon \Big(\|u_t\|^2+\|v_t\|^2- \|\nabla^\kappa u\|^2- \|\nabla^\kappa v\|^2+\|uv\|_p^p-G(t)\Big)
 \\
 \geq&   \bigg((1-\nu)\mathcal{K}^{-\nu}(t)-\frac{1}{4}\epsilon \varpi \bigg)\mathcal{K}'(t)
+\epsilon \big(\|u_t\|^2+\|v_t\|^2- \|\nabla^\kappa u\|^2- \|\nabla^\kappa v\|^2+\|uv\|_p^p \big)
 \\
 -&  \frac{\varpi^{-\frac{2+l}{l}}  }{4} \big(\|u\|^{l+2}+\|v\|^{l+2}\big).\label{5.7}
\end{aligned}
\end{equation}
We can choose the Young parameter $\varpi$ as a time–dependent weight of $\mathcal K(t)$, namely
$$
\varpi = k\,\mathcal K^{-\nu}(t), \qquad k>0 .
$$
Moreover,
\begin{equation*}
\varpi^{-\frac{l+2}{l}}
 = k^{-\frac{l+2}{l}}\,
\mathcal K^{\frac{\nu(l+2)}{l}}(t),
\end{equation*}
where the constant $k$ is redefined if necessary.


Substituting these estimates into \eqref{5.7}, we obtain
\begin{align*}
z'(t)\ge &
\Big(1-\nu-\epsilon\frac{k}{4}\Big)
\mathcal K^{-\nu}(t)\mathcal K'(t)
\\
&
+\epsilon\bigg(
\|u_t\|^2+\|v_t\|^2
-\|\nabla^\kappa u\|^2-\|\nabla^\kappa v\|^2
+\|uv\|_p^p
-\frac{1}{4k^{\frac{l+2}{l}}}\mathcal K^{ \frac{\nu(l+2)}{l}}(t)
\big(\|u\|^{l+2}+\|v\|^{l+2}\big)
\bigg),
\end{align*}
We take
\begin{equation} \label{alpha}
0 \le \alpha < \frac{p-2}{p}.
\end{equation}
Then,
\begin{equation}
\begin{aligned}
 z'(t)
\geq& \bigg(1-\nu-\epsilon  \frac{k}{4}\bigg)
\mathcal{K}^{-\nu}(t)\mathcal{K}'(t)+\epsilon p(1-\alpha)\mathcal{K}(t)+\epsilon \bigg(\frac{ p(1-\alpha)}{2}+1\bigg) (\|u_t\|^2+\|v_t\|^2)
 \\
+&\epsilon \left(\frac{ p(1-\alpha)}{2}-1\right) \big(\|\nabla^\kappa u\|^2+\|\nabla^\kappa v\|^2\big)
+\epsilon p\|uv\|_p^p
\\
&- \epsilon \frac{1}{4k^{\frac{l+2}{l}}}\mathcal K^{ \frac{\nu(l+2)}{l}}(t) \big(\|u\|^{l+2}+\|v\|^{l+2}\big).\label{5.8}
\end{aligned}
\end{equation}
As $ l>0$,  choosing $\nu$ so that $$0<\nu< \frac{l}{l+2},$$
and using the upper bound in \eqref{5.3}, we obtain
$$
\mathcal K^{\frac{\nu(l+2)}{l}}(t)\leq\mathcal K(t)
\le \frac{1}{p}\|uv\|_p^p,
$$
and hence
\begin{equation*}
 \epsilon \frac{1}{4k^{\frac{l+2}{l}}}\,
\mathcal K^{\frac{\nu(l+2)}{l}}(t)\big(\|u\|^{l+2}+\|v\|^{l+2}\big)
   \le
\epsilon \frac{1}{4k^{\frac{l+2}{l}}p}\,
\|uv\|_p^p
\big(\|u\|^{l+2}+\|v\|^{l+2}\big).
\end{equation*}
Since $$\|u\|^{l+2}+\|v\|^{l+2}\ge0,$$ it follows immediately that
$$
\epsilon \frac{1}{4k^{\frac{l+2}{l}}}\,
\mathcal K^{\frac{\nu(l+2)}{l}}(t)
\big(\|u\|^{l+2}+\|v\|^{l+2}\big)
\;\le\;
\epsilon \frac{1}{4k^{\frac{l+2}{l}}p}\,\|uv\|_p^p.
$$
Then
\begin{equation}
\begin{aligned}
z'(t)
\geq&\left (1-\nu-\epsilon  \frac{k}{4}\right)
\mathcal{K}^{-\nu}(t)\mathcal{K}'(t)+\epsilon p(1-\alpha) \mathcal{K}(t)+\epsilon\left(\frac{ p(1-\alpha)}{2}+1\right) (\|u_t\|^2+\|v_t\|^2)\label{5.9}\\
& +\epsilon\left(\frac{ p(1-\alpha)}{2}-1 \right) (\|\nabla^\kappa u\|^2+\|\nabla^\kappa v\|^2)
+\epsilon \left(p- \epsilon \frac{1}{4k^{\frac{l+2}{l}}p}\right)\|uv\|_p^p.
\end{aligned}
\end{equation}
Choosing $\alpha$ and $k$ such that
\begin{align}
&p(1-\alpha) >2, \nonumber
\\
 \label{k}
k>\left(\frac{\epsilon}{4p^2}\right)^{\frac{l}{l+2}},
\end{align}
and then $\epsilon$ such that
$$
1-\nu-\epsilon  \frac{k}{4}\geq 0,
$$
and we have
$$
z(t)=\mathcal{K}^{1-\nu}(0)+\epsilon \int\limits_{\Omega} [u_{1}u_0+v_{1}v_0]\,dx> 0.$$
We can pick
$$
\nu \in \Big(0, \frac{l}{l+2}\Big), \quad \alpha \in \Big[0, \frac{p-2}{p}\Big), \quad k \in \Big( (\tfrac{\epsilon}{4p^2})^{\frac{l}{l+2}}, \frac{4(1-\nu)}{\epsilon} \Big]
$$
so that we can choose parameters satisfying all conditions simultaneously.
Then \eqref{5.9} becomes
\begin{align}
&z'(t)\ge \epsilon K \left(\mathcal{K}(t)+ \|u_t\|^2+\|v_t\|^2+\|uv\|_p^p\right)\ge 0,\qquad K>0.\label{5.10}
\end{align}
The functional $z(t)$ is increasing on $(0, t_{\mathrm{max}})$  with
$$z(t)\ge z(0)>0\quad \text{for all}\quad t\ge 0.$$
We  estimate $z^{\frac{1}{1-\nu}}(t)$ by using the elementary inequality
$$| \alpha + \beta |^s \le 2^{s-1}( | \alpha |^s+ | \beta |^s) ,\quad s\ge 1,$$
and the Young's inequality
\begin{equation}
\begin{aligned}
&z^{\frac{1}{1-\nu}}(t)\le 2^{\frac{1}{1-\nu}}\left(\mathcal{K}(t)
+\epsilon \|u\|^{\frac{1}{1-\nu}}\|u_t\|^{\frac{1}{1-\nu}}+\epsilon \|v\|^{\frac{1}{1-\nu}}\|v_t\|^{\frac{1}{1-\nu}}\right)
\\
&\le K\left(\mathcal{K}(t)+\|u_t\|^{2}+\|v_t\|^{2}+\|u\|^{\frac{2}{1-2\nu}} +\|v\|^{\frac{2}{1-2\nu}}\right).
\end{aligned}\label{5.11}
\end{equation}
Since $\nu<\frac{1}{2}-\frac{1}{p}$, using the inequality
$$\xi^\tau \le \Big(1 +\frac{1}{s}\Big)(s + \xi),\quad \xi \ge 0,\quad 0 \le\tau \le 1,\quad s > 0,$$
and taking $$\xi=\|u\|_2^p, \qquad \tau=\frac{2}{(1-2\nu)p}<1,$$  and $s=\mathcal{K}(0)$.
We get, for some constants $K_1, K_2>0$,
\begin{equation}
\begin{aligned}
\|u\|^{\frac{2}{1-2\nu}}+\|v\|^{\frac{2}{1-2\nu}} \leq& \left(1 +\frac{1}{\mathcal{K}(t=0)}\right)\left(\mathcal{K}(0) + \|u\|_2^p \right)+ \left(1 +\frac{1}{\mathcal{K}(t=0)}\right)\left(\mathcal{K}(0) + \|v\|_2^p \right) \\
 \leq& K_1\left(\mathcal{K}(t) + \|u\|_p^p+ \|v\|_p^p\right)
 \leq  K_2\left(\mathcal{K}(t) + \sigma\|u v\|_p^p+C(\sigma)
\right), \quad \sigma>0.\label{5.12}
\end{aligned}
\end{equation}
Substituting \eqref{5.12} into \eqref{5.11}, we find
\begin{align}
&z^{\frac{1}{1-\nu}}(t)\le K_3\left(\mathcal{K}(t)+\|u_t\|^{2}+\|v_t\|^{2}+\|uv\|_p^{p}\right), \quad K_3>0.\label{5.13}
\end{align}
By \eqref{5.10}, \eqref{5.13}, we get
\begin{align}
&z'(t)\ge K_4 z^{\frac{1}{1-\nu}}(t), K_4>0.\label{5.14}
\end{align}
A simple integration of \eqref{5.14} over $(0, t)$ then yields
$$z(t)\ge \left(z^{\frac{-\nu}{1-\nu}}(t=0)-K_4\frac{\nu t}{1-\nu}\right)^{\frac{\nu-1}{\nu}}.$$
Thus, there exists $t_2$ such that
\begin{align}
&t_2\le t_*=\frac{z^{\frac{-\nu}{1-\nu}}(t=0)(1-\nu)}{K_4\nu}, \nonumber
\\
&\lim_{t\rightarrow t_2^{-}}z(t)=+\infty.\label{5.15}
\end{align}
The proof is complete.
\end{proof}

\vskip10pt

\subsection{Positive initial energy}
Let
\begin{align}
&0\le E(t=0)\le E_1,\nonumber
\\
&E_{1}
=\left(\frac{1}{2}-\frac{1}{p}\right)C_0^{-\frac{2p}{p-2}}, \label{5.16}
\end{align}
and $C_0$ is the best Poincar\'e  constant.

\begin{lemma}\label{lemma3.5}
    \cite{Wu2009TJM} If $$0\le E(t=0)< E_1, \qquad \|\nabla^\kappa u_0\|+\|\nabla^\kappa v_0\|>\lambda_1,$$
 then there exists $\lambda_2>\lambda_1$ such  that $$\|\nabla^\kappa u\|+\|\nabla^\kappa v\|\ge \lambda_2, \qquad \lambda_1=C_0^{-\frac{p}{p-2}}.$$
\end{lemma}

\begin{theorem} \label{theoremblowup2}Under the assumptions of Theorem~\ref{Theoremlocal}, for  $$0\le E(0)< E_1,\qquad \|\nabla^\kappa u_0\|+\|\nabla^\kappa v_0\|> \lambda_1,$$
 the solution to problem
\eqref{1.1}--\eqref{1.3} blows up, that is, $t_{\mathrm{max}}<+\infty$.
\end{theorem}
\begin{proof}

We set $$\mathcal{K}(t)=E_2-E(t),\qquad E_2=\frac{E(t=0)+E_1}{2}.$$ Proceeding as in \eqref{5.8}  with $\alpha=0$, we obtain
\begin{equation}
\begin{aligned}
z'(t)\ge &\left(1-\nu-\epsilon  \frac{k}{4}\right)
\mathcal{K}^{-\nu}(t)\mathcal{K}'(t)+\epsilon p\mathcal{K}(t)+\epsilon\left(\frac{ p}{2}+1\right) (\|u_t\|^2+\|v_t\|^2)
\\
&+\epsilon\left(\frac{ p}{2}-1\right) (\|\nabla^\kappa u\|^2+\|\nabla^\kappa v\|^2)
+\epsilon p\|uv\|_p^p
\\
&- \frac{\epsilon}{4k^{\frac{l+2}{l}}}\mathcal K^{ \frac{\nu(l+2)}{l}}(t) (\|u\|^{l+2}+\|v\|^{l+2}) -\epsilon pE_2.
\end{aligned}\label{5.17}
\end{equation}
Moreover,
\begin{equation}
\begin{aligned}
\epsilon\left(\frac{ p}{2}-1\right) (\|\nabla^\kappa u\|^2+\|\nabla^\kappa v\|^2) -\epsilon pE_2
=&\epsilon \left(\left(\frac{ p}{2}-1\right) (\|\nabla^\kappa u\|^2+\|\nabla^\kappa v\|^2) -pE_2\right)
\\
=&\epsilon\Big(\left(\frac{ p}{2}-1\right)\frac{\lambda_2^2-\lambda_1^2}{\lambda_2^2} (\|\nabla^\kappa u\|^2+\|\nabla^\kappa v\|^2) \\
&+\left(\frac{ p}{2}-1\right)\lambda_1^2\frac{\|\nabla^\kappa u\|^2+\|\nabla^\kappa v\|^2}{\lambda_2^2} -pE_2\Big) \\
\ge &\epsilon \left(K_5(\|\nabla^\kappa u\|^2+\|\nabla^\kappa v\|^2)+K_6\right),
\end{aligned}\label{5.18}
\end{equation}
where $\lambda_2$ is given in Lemma \ref{lemma3.5},
$$K_5=\left(\frac{ p}{2}-1\right)\frac{\lambda_2^2-\lambda_1^2}{\lambda_2^2}>0,\qquad K_6=\left(\frac{ p}{2}-1\right)\lambda_1^2 -pE_2.$$
By \eqref{5.16} and Lemma  \ref{lemma3.5}, we observe that
\begin{equation} K_6=\left(\frac{ p}{2}-1\right)\lambda_1^2 -\frac{p\left(E_1+E(t=0)\right)}{2} =\frac{p\left(E_1-E(t=0)\right)}{2} >0.
 \label{5.19}
\end{equation}
Thus, by \eqref{5.18},\quad \eqref{5.19} and \eqref{5.17}, we arrive at
\begin{equation}
\begin{aligned}
z'(t)\ge& \left(1-\nu-\epsilon  \frac{k}{4}\right)
\mathcal{K}^{-\nu}(t)\mathcal{K}'(t)+\epsilon p\mathcal{K}(t)+\epsilon\left(\frac{ p}{2}+1\right) (\|u_t\|^2+\|v_t\|^2)\nonumber\\
&+\epsilon K_5 (\|\nabla^\kappa u\|^2+\|\nabla^\kappa v\|^2)
+\epsilon p\|uv\|_p^p-\epsilon \frac{1}{4k^{\frac{l+2}{l}}}\mathcal K^{ \frac{\nu(l+2)}{l}}(t) (\|u\|^{l+2}+\|v\|^{l+2}).
\end{aligned}\label{5.20}
\end{equation}
Similarly to  \eqref{5.12}  we have
\begin{align*}
z'(t)\ge &\left(1-\nu-\epsilon  \frac{k}{4}\right)
\mathcal{K}^{-\nu}(t)\mathcal{K}'(t)+\epsilon p\mathcal{K}(t)+\epsilon\left(\frac{ p}{2}+1\right) (\|u_t\|^2+\|v_t\|^2)
\\
&+\epsilon K_{5}  (\|\nabla^\kappa u\|^2+\|\nabla^\kappa v\|^2)
+\epsilon \left(p-\epsilon \frac{\sigma}{4k^{\frac{l+2}{l}}p}\right)\|uv\|_p^p+C(\sigma).
\end{align*}
Choosing $k$  as in \eqref{k},  we can complete the proof as in Theorem \ref{theoremblowup}.
\end{proof}

\vskip30pt



\end{document}


